Author’s response to reviews

Title: Impact of molar teeth distalization with clear aligners on occlusal vertical dimension: A retrospective study

Authors:
Silvia Caruso (silvia.caruso21@gmail.com)
Alessandro Nota (dr.alessandro.nota@gmail.com)
Shideh Ehsani (shideh_ehsani@yahoo.it)
Elena Maddalone (elemadda93@hotmail.it)
Kenji Ojima (simtecc@tin.it)
Simona Tecco (simtecc@gmail.com)

Version: 2 Date: 05 Aug 2019

Author’s response to reviews:

Reviewer Comments for Manuscript
OHEA-D-19-00216R1
"Impact of molar teeth distalization with orthodontic aligners on occlusal vertical dimension: a retrospective study"

Editor Comments:

Dear Authors, please resubmit your paper after having modified it according to the last request of reviewer n° 2.

R:Dear Editor, we are resubmitting the paper after having done the last request on the reviewer n°2, we thank you for your availability and time and we hope that this last version of the manuscript reached now the quality level for being published on this important journal.

BMC Oral Health operates a policy of open peer review, which means that you will be able to see the names of the reviewers who provided the reports via the online peer review system. We
encourage you to also view the reports there, via the action links on the left-hand side of the page, to see the names of the reviewers.

Reviewer reports:

Alessandro Mangano, D.D.S., M.S. (Reviewer 1): The article is well structured and the statistical analysis is satisfactory. The limitations of the study are clearly indicated and explained. The study reported interesting data that should be investigated with further longitudinal studies.

R: We thank the reviewer for his time and availability that helped us to improve the quality of our manuscript.

Suleyman Kutalmış Buyuk, D.D.S., Ph.D. (Reviewer 2): The authors have successfully addressed previously raised comments.

Only, the authors should mention the exact values of n.s. values in Table 1.

R: According to the request we added to Table 1 all the exact p-values for non significant variables.

We thank the reviewer for his time and availability that helped us to improve the quality of our manuscript.

Best Regards,

The Authors